Author's response to reviews

Title: Nurse led, primary care based antiretroviral treatment versus hospital care: a controlled prospective study in Swaziland

Authors:

Ciaran P Humphreys (ciaranhumphreys@gmail.com)
John Wright (john.wright@bradfordhospitals.nhs.uk)
John Walley (j.walley@leeds.ac.uk)
Canaan T Mamvura (docmamvura@yahoo.com)
Kerry A Bailey (kerrybaileyjones@googlemail.com)
Sweetness N Ntshalintshali (docmamvura@yahoo.com)
Robert M West (r.m.west@leeds.ac.uk)
Aby Philip (abysloba@gmail.com)

Version: 2 Date: 15 January 2010

Author's response to reviews:

Dear editor,

Kindly find attached a revised version of the manuscript. The revision is on page 8, in relation to consent, with the addition of the following text:

"This was an evaluation of planned service change. Consent for transfer to nurse led primary care based antiretroviral treatment was sought from those in the intervention group. Consent was sought verbally by the assessing physician and recorded by the physician in a written proforma. No personally identifiable data was shared beyond the clinical team."